% bare_jrnl_transmag.tex
%% V1.4
%% 2012/12/27
%% by Michael Shell
%% see http://www.michaelshell.org/
%% for current contact information.
%%
%% This is a skeleton file demonstrating the use of IEEEtran.cls
%% (requires IEEEtran.cls version 1.8 or later) with an IEEE
%% Transactions on Magnetics journal paper.
%%
%% Support sites:
%% http://www.michaelshell.org/tex/ieeetran/
%% http://www.ctan.org/tex-archive/macros/latex/contrib/IEEEtran/
%% and
%% http://www.ieee.org/



% *** Authors should verify (and, if needed, correct) their LaTeX system  ***
% *** with the testflow diagnostic prior to trusting their LaTeX platform ***
% *** with production work. IEEE's font choices can trigger bugs that do  ***
% *** not appear when using other class files.                            ***
% The testflow support page is at:
% http://www.michaelshell.org/tex/testflow/


%%*************************************************************************
%% Legal Notice:
%% This code is offered as-is without any warranty either expressed or
%% implied; without even the implied warranty of MERCHANTABILITY or
%% FITNESS FOR A PARTICULAR PURPOSE!
%% User assumes all risk.
%% In no event shall IEEE or any contributor to this code be liable for
%% any damages or losses, including, but not limited to, incidental,
%% consequential, or any other damages, resulting from the use or misuse
%% of any information contained here.
%%
%% All comments are the opinions of their respective authors and are not
%% necessarily endorsed by the IEEE.
%%
%% This work is distributed under the LaTeX Project Public License (LPPL)
%% ( http://www.latex-project.org/ ) version 1.3, and may be freely used,
%% distributed and modified. A copy of the LPPL, version 1.3, is included
%% in the base LaTeX documentation of all distributions of LaTeX released
%% 2003/12/01 or later.
%% Retain all contribution notices and credits.
%% ** Modified files should be clearly indicated as such, including  **
%% ** renaming them and changing author support contact information. **
%%
%% File list of work: IEEEtran.cls, IEEEtran_HOWTO.pdf, bare_adv.tex,
%%                    bare_conf.tex, bare_jrnl.tex, bare_jrnl_compsoc.tex,
%%                    bare_jrnl_transmag.tex
%%*************************************************************************

% Note that the a4paper option is mainly intended so that authors in
% countries using A4 can easily print to A4 and see how their papers will
% look in print - the typesetting of the document will not typically be
% affected with changes in paper size (but the bottom and side margins will).
% Use the testflow package mentioned above to verify correct handling of
% both paper sizes by the user's LaTeX system.
%
% Also note that the "draftcls" or "draftclsnofoot", not "draft", option
% should be used if it is desired that the figures are to be displayed in
% draft mode.
%
\documentclass[journal,transmag]{IEEEtran}
%
% If IEEEtran.cls has not been installed into the LaTeX system files, manually specify the path to it like:
% \documentclass[journal]{../sty/IEEEtran}

% Some very useful LaTeX packages include: (uncomment the ones you want to load)

% *** MISC UTILITY PACKAGES ***
%
%\usepackage{ifpdf}
% Heiko Oberdiek's ifpdf.sty is very useful if you need conditional
% compilation based on whether the output is pdf or dvi.
% usage:
% \ifpdf
%   % pdf code
% \else
%   % dvi code
% \fi
% The latest version of ifpdf.sty can be obtained from:
% http://www.ctan.org/tex-archive/macros/latex/contrib/oberdiek/
% Also, note that IEEEtran.cls V1.7 and later provides a builtin
% \ifCLASSINFOpdf conditional that works the same way.
% When switching from latex to pdflatex and vice-versa, the compiler may
% have to be run twice to clear warning/error messages.

% *** CITATION PACKAGES ***
%
%\usepackage{cite}
% cite.sty was written by Donald Arseneau
% V1.6 and later of IEEEtran pre-defines the format of the cite.sty package
% \cite{} output to follow that of IEEE. Loading the cite package will
% result in citation numbers being automatically sorted and properly
% "compressed/ranged". e.g., [1], [9], [2], [7], [5], [6] without using
% cite.sty will become [1], [2], [5]--[7], [9] using cite.sty. cite.sty's
% \cite will automatically add leading space, if needed. Use cite.sty's
% noadjust option (cite.sty V3.8 and later) if you want to turn this off
% such as if a citation ever needs to be enclosed in parenthesis.
% cite.sty is already installed on most LaTeX systems. Be sure and use
% version 4.0 (2003-05-27) and later if using hyperref.sty. cite.sty does
% not currently provide for hyperlinked citations.
% The latest version can be obtained at:
% http://www.ctan.org/tex-archive/macros/latex/contrib/cite/
% The documentation is contained in the cite.sty file itself.

% *** GRAPHICS RELATED PACKAGES ***
%
\ifCLASSINFOpdf
  \usepackage[pdftex]{graphicx}
  % declare the path(s) where your graphic files are
  \graphicspath{{../pdf/}{../jpeg/}}
  % and their extensions so you won't have to specify these with
  % every instance of \includegraphics
  \DeclareGraphicsExtensions{.pdf,.jpeg,.png}
\else
  % or other class option (dvipsone, dvipdf, if not using dvips). graphicx
  % will default to the driver specified in the system graphics.cfg if no
  % driver is specified.
  \usepackage[dvips]{graphicx}
  % declare the path(s) where your graphic files are
  \graphicspath{{../eps/}}
  % and their extensions so you won't have to specify these with
  % every instance of \includegraphics
  \DeclareGraphicsExtensions{.eps}
\fi
% graphicx was written by David Carlisle and Sebastian Rahtz. It is
% required if you want graphics, photos, etc. graphicx.sty is already
% installed on most LaTeX systems. The latest version and documentation
% can be obtained at:
% http://www.ctan.org/tex-archive/macros/latex/required/graphics/
% Another good source of documentation is "Using Imported Graphics in
% LaTeX2e" by Keith Reckdahl which can be found at:
% http://www.ctan.org/tex-archive/info/epslatex/
%
% latex, and pdflatex in dvi mode, support graphics in encapsulated
% postscript (.eps) format. pdflatex in pdf mode supports graphics
% in .pdf, .jpeg, .png and .mps (metapost) formats. Users should ensure
% that all non-photo figures use a vector format (.eps, .pdf, .mps) and
% not a bitmapped formats (.jpeg, .png). IEEE frowns on bitmapped formats
% which can result in "jaggedy"/blurry rendering of lines and letters as
% well as large increases in file sizes.
%
% You can find documentation about the pdfTeX application at:
% http://www.tug.org/applications/pdftex

% *** MATH PACKAGES ***
%
%\usepackage[cmex10]{amsmath}
% A popular package from the American Mathematical Society that provides
% many useful and powerful commands for dealing with mathematics. If using
% it, be sure to load this package with the cmex10 option to ensure that
% only type 1 fonts will utilized at all point sizes. Without this option,
% it is possible that some math symbols, particularly those within
% footnotes, will be rendered in bitmap form which will result in a
% document that can not be IEEE Xplore compliant!
%
% Also, note that the amsmath package sets \interdisplaylinepenalty to 10000
% thus preventing page breaks from occurring within multiline equations. Use:
%\interdisplaylinepenalty=2500
% after loading amsmath to restore such page breaks as IEEEtran.cls normally
% does. amsmath.sty is already installed on most LaTeX systems. The latest
% version and documentation can be obtained at:
% http://www.ctan.org/tex-archive/macros/latex/required/amslatex/math/

% *** SPECIALIZED LIST PACKAGES ***
%
%\usepackage{algorithmic}
% algorithmic.sty was written by Peter Williams and Rogerio Brito.
% This package provides an algorithmic environment fo describing algorithms.
% You can use the algorithmic environment in-text or within a figure
% environment to provide for a floating algorithm. Do NOT use the algorithm
% floating environment provided by algorithm.sty (by the same authors) or
% algorithm2e.sty (by Christophe Fiorio) as IEEE does not use dedicated
% algorithm float types and packages that provide these will not provide
% correct IEEE style captions. The latest version and documentation of
% algorithmic.sty can be obtained at:
% http://www.ctan.org/tex-archive/macros/latex/contrib/algorithms/
% There is also a support site at:
% http://algorithms.berlios.de/index.html
% Also of interest may be the (relatively newer and more customizable)
% algorithmicx.sty package by Szasz Janos:
% http://www.ctan.org/tex-archive/macros/latex/contrib/algorithmicx/

% *** ALIGNMENT PACKAGES ***
%
%\usepackage{array}
% Frank Mittelbach's and David Carlisle's array.sty patches and improves
% the standard LaTeX2e array and tabular environments to provide better
% appearance and additional user controls. As the default LaTeX2e table
% generation code is lacking to the point of almost being broken with
% respect to the quality of the end results, all users are strongly
% advised to use an enhanced (at the very least that provided by array.sty)
% set of table tools. array.sty is already installed on most systems. The
% latest version and documentation can be obtained at:
% http://www.ctan.org/tex-archive/macros/latex/required/tools/

% IEEEtran contains the IEEEeqnarray family of commands that can be used to
% generate multiline equations as well as matrices, tables, etc., of high
% quality.

% *** SUBFIGURE PACKAGES ***
%\ifCLASSOPTIONcompsoc
%  \usepackage[caption=false,font=normalsize,labelfont=sf,textfont=sf]{subfig}
%\else
%  \usepackage[caption=false,font=footnotesize]{subfig}
%\fi
% subfig.sty, written by Steven Douglas Cochran, is the modern replacement
% for subfigure.sty, the latter of which is no longer maintained and is
% incompatible with some LaTeX packages including fixltx2e. However,
% subfig.sty requires and automatically loads Axel Sommerfeldt's caption.sty
% which will override IEEEtran.cls' handling of captions and this will result
% in non-IEEE style figure/table captions. To prevent this problem, be sure
% and invoke subfig.sty's "caption=false" package option (available since
% subfig.sty version 1.3, 2005/06/28) as this is will preserve IEEEtran.cls
% handling of captions.
% Note that the Computer Society format requires a larger sans serif font
% than the serif footnote size font used in traditional IEEE formatting
% and thus the need to invoke different subfig.sty package options depending
% on whether compsoc mode has been enabled.
%
% The latest version and documentation of subfig.sty can be obtained at:
% http://www.ctan.org/tex-archive/macros/latex/contrib/subfig/



% *** FLOAT PACKAGES ***
%
%\usepackage{fixltx2e}
% fixltx2e, the successor to the earlier fix2col.sty, was written by
% Frank Mittelbach and David Carlisle. This package corrects a few problems
% in the LaTeX2e kernel, the most notable of which is that in current
% LaTeX2e releases, the ordering of single and double column floats is not
% guaranteed to be preserved. Thus, an unpatched LaTeX2e can allow a
% single column figure to be placed prior to an earlier double column
% figure. The latest version and documentation can be found at:
% http://www.ctan.org/tex-archive/macros/latex/base/


%\usepackage{stfloats}
% stfloats.sty was written by Sigitas Tolusis. This package gives LaTeX2e
% the ability to do double column floats at the bottom of the page as well
% as the top. (e.g., "\begin{figure*}[!b]" is not normally possible in
% LaTeX2e). It also provides a command:
%\fnbelowfloat
% to enable the placement of footnotes below bottom floats (the standard
% LaTeX2e kernel puts them above bottom floats). This is an invasive package
% which rewrites many portions of the LaTeX2e float routines. It may not work
% with other packages that modify the LaTeX2e float routines. The latest
% version and documentation can be obtained at:
% http://www.ctan.org/tex-archive/macros/latex/contrib/sttools/
% Do not use the stfloats baselinefloat ability as IEEE does not allow
% \baselineskip to stretch. Authors submitting work to the IEEE should note
% that IEEE rarely uses double column equations and that authors should try
% to avoid such use. Do not be tempted to use the cuted.sty or midfloat.sty
% packages (also by Sigitas Tolusis) as IEEE does not format its papers in
% such ways.
% Do not attempt to use stfloats with fixltx2e as they are incompatible.
% Instead, use Morten Hogholm'a dblfloatfix which combines the features
% of both fixltx2e and stfloats:
%
% \usepackage{dblfloatfix}
% The latest version can be found at:
% http://www.ctan.org/tex-archive/macros/latex/contrib/dblfloatfix/




%\ifCLASSOPTIONcaptionsoff
%  \usepackage[nomarkers]{endfloat}
% \let\MYoriglatexcaption\caption
% \renewcommand{\caption}[2][\relax]{\MYoriglatexcaption[#2]{#2}}
%\fi
% endfloat.sty was written by James Darrell McCauley, Jeff Goldberg and
% Axel Sommerfeldt. This package may be useful when used in conjunction with
% IEEEtran.cls'  captionsoff option. Some IEEE journals/societies require that
% submissions have lists of figures/tables at the end of the paper and that
% figures/tables without any captions are placed on a page by themselves at
% the end of the document. If needed, the draftcls IEEEtran class option or
% \CLASSINPUTbaselinestretch interface can be used to increase the line
% spacing as well. Be sure and use the nomarkers option of endfloat to
% prevent endfloat from "marking" where the figures would have been placed
% in the text. The two hack lines of code above are a slight modification of
% that suggested by in the endfloat docs (section 8.4.1) to ensure that
% the full captions always appear in the list of figures/tables - even if
% the user used the short optional argument of \caption[]{}.
% IEEE papers do not typically make use of \caption[]'s optional argument,
% so this should not be an issue. A similar trick can be used to disable
% captions of packages such as subfig.sty that lack options to turn off
% the subcaptions:
% For subfig.sty:
% \let\MYorigsubfloat\subfloat
% \renewcommand{\subfloat}[2][\relax]{\MYorigsubfloat[]{#2}}
% However, the above trick will not work if both optional arguments of
% the \subfloat command are used. Furthermore, there needs to be a
% description of each subfigure *somewhere* and endfloat does not add
% subfigure captions to its list of figures. Thus, the best approach is to
% avoid the use of subfigure captions (many IEEE journals avoid them anyway)
% and instead reference/explain all the subfigures within the main caption.
% The latest version of endfloat.sty and its documentation can obtained at:
% http://www.ctan.org/tex-archive/macros/latex/contrib/endfloat/
%
% The IEEEtran \ifCLASSOPTIONcaptionsoff conditional can also be used
% later in the document, say, to conditionally put the References on a
% page by themselves.




% *** PDF, URL AND HYPERLINK PACKAGES ***
%
%\usepackage{url}
% url.sty was written by Donald Arseneau. It provides better support for
% handling and breaking URLs. url.sty is already installed on most LaTeX
% systems. The latest version and documentation can be obtained at:
% http://www.ctan.org/tex-archive/macros/latex/contrib/url/
% Basically, \url{my_url_here}.




% *** Do not adjust lengths that control margins, column widths, etc. ***
% *** Do not use packages that alter fonts (such as pslatex).         ***
% There should be no need to do such things with IEEEtran.cls V1.6 and later.
% (Unless specifically asked to do so by the journal or conference you plan
% to submit to, of course. )


% correct bad hyphenation here
\hyphenation{op-tical net-works semi-conduc-tor}


\begin{document}
\begin{thebibliography}{1}

\bibitem{VARIUM13}
  Farias,~M.~C.~Q.; Heynderickx,~I.; Bruno,~L.~M.~E.; Redi,~J.~A., \emph{Visual Artifacts Interference Understanding and Modeling (VARIUM)}. In: Seventh International Workshop on Video Processing and Quality Metrics for Consumer Electronics, 2013, Scottsdale.

\bibitem{ISCAS13}
  Redi,~J.~A.; Heynderickx,~I.; Bruno,~L.~M.~E.; Farias,~M.~C.~Q., \emph{On the impact of packet-loss impairments on visual attention mechanisms}. In: ISCAS, pp. 1107-1110. IEEE, 2013.

\bibitem{ICME03}
  Farias,~M.~C.~Q.; Foley,~J.~M.; Mitra,~S.~K., \emph{Perceptual contributions of blocky, blurry and noisy artifacts to overall annoyance}. In: Proc. IEEE International Conference on Multimedia \& Expo, Baltimore, MD, USA, 2003, Vol. 1, pp. 529–532.

\bibitem{SPIE03}
  Farias,~M.~C.~Q.; Foley,~J.~M.; Mitra,~S.~K., \emph{Some Properties of Synthetic Blocky and Blurry Artifacts}. In: Proceedings of the SPIE, Human Vision and Electronic Imaging VIII, Vol. 5007, pp. 128-136 (2003).

\bibitem{SPIE05}
  Farias,~M.~C.~Q.; Foley,~J.~M.; Mitra,~S.~K., \emph{Perceptual analysis of video impairments that combine blocky, blurry, noisy, and ringing synthetic artifacts}. In: Proceedings of the SPIE, Human Vision and Electronic Imaging VIII, Vol. 5666, pp. 107-118 (2005).

\bibitem{ICASSP04}
    Farias,~M.~C.~Q.; Foley,~J.~M.; Mitra,~S.~K., \emph{Detectability and Annoyance of Synthetic Blurring and Ringing in Video Sequences}. In: International Conference on Acoustics, Speech, and Signal Processing - ICASSP, Vol. 3, pp. 697-700, 2004.

\bibitem{SSDT02}
  Farias,~M.~C.~Q.; Foley,~J.~M.; Mitra,~S.~K., \emph{Detectability and Annoyance of Synthetic Blocky and Blurry Video Artifacts}. In: Journal Sid Symposium Digest of Technical Papers, Vol. 33, No. 1, 2002.

\bibitem{ICASSP05}
  Farias,~M.~C.~Q.; Foley,~J.~M.; Mitra,~S.~K., \emph{Detectability and annoyance of synthetic blockiness, blurriness, noisiness, and ringing in video sequences}. In: Proc. IEEE International Conference on Acoustics, Speech, and Signal Processing, Philadelphia, PA, Vol. 2, pp. 553–556, 2005.

\bibitem{ITU98}
  International Telecommunication Union, \emph{ITU-T Recommendation BT.500-8: Methodology for the subjective assessment of the quality of television pictures}. 1998.

\bibitem{ICIP05}
  Farias,~M.~C.~Q.; Mitra,~S.~K., \emph{No-reference video quality metric based on artifact measurements}. In: ICIP, IEEE, pp. 141-145, 2005.

\bibitem{SPSS04}
  Landau,~S.; Everitt,~B.~S., \emph{A Handbook of Statistical Analyses Using SPSS}. ISBN: 9781584883692, Statistics Chapman \& Hall/CRC, Taylor \& Francis, 2004.

\bibitem{Statist09}
  Montgomery,~D.,~C., \emph{Introduction To Statistical Quality Control}. ISBN: 9780470233979, John Wiley \& Sons, Incorporated, 2009.

\bibitem{IMGPRO10}
  Seshadrinathan,~K.; Soundararajan,~R.; Bovik,~A.~C.; Cormack,~L.~K., \emph{Study of Subjective and Objective Quality Assessment of Video}. In: IEEE Transactions on Image Processing, Vol. 19, No. 6, Piscataway, NJ, USA, 2010.
\end{thebibliography}

%
% paper title
% can use linebreaks \\ within to get better formatting as desired
% Do not put math or special symbols in the title.
\title{REPORT: Analysis of the influence of different artifacts in the video content from three experiments.}

% author names and affiliations
% transmag papers use the long conference author name format.

\author{\IEEEauthorblockN{Alexandre Fieno da Silva\IEEEauthorrefmark{1},
Judith A. Redi\IEEEauthorrefmark{2}, and
Myl\`{e}ne C.Q. Farias\IEEEauthorrefmark{1}}
\IEEEauthorblockA{\IEEEauthorrefmark{1}University of Brasilia, Campus Universitario Darcy Ribeiro, Brasilia - DF, Brazil, 70919-970}
\IEEEauthorblockA{\IEEEauthorrefmark{2}Delft University of Technology, Mekelweg 4, Delft, The Netherlands, 2628 CD}}
%%%%\thanks{Manuscript received December 1, 2012; revised December 27, 2012.
%%%%Corresponding author: A. F. Silva (email: alexandrefieno@gmail.com).}}

% The paper headers
%%%%\markboth{Journal of \LaTeX\ Class Files,~Vol.~11, No.~4, December~2012}%
%%%%%{Shell \MakeLowercase{\textit{et al.}}: Bare Demo of IEEEtran.cls for Journals}
% The only time the second header will appear is for the odd numbered pages
% after the title page when using the twoside option.
%
% *** Note that you probably will NOT want to include the author's ***
% *** name in the headers of peer review papers.                   ***
% You can use \ifCLASSOPTIONpeerreview for conditional compilation here if
% you desire.


% If you want to put a publisher's ID mark on the page you can do it like
% this:
%\IEEEpubid{0000--0000/00\$00.00~\copyright~2012 IEEE}
% Remember, if you use this you must call \IEEEpubidadjcol in the second
% column for its text to clear the IEEEpubid mark.



% use for special paper notices
%\IEEEspecialpapernotice{(Invited Paper)}


% for Transactions on Magnetics papers, we must declare the abstract and
% index terms PRIOR to the title within the \IEEEtitleabstractindextext
% IEEEtran command as these need to go into the title area created by
% \maketitle.
% As a general rule, do not put math, special symbols or citations
% in the abstract or keywords.
\IEEEtitleabstractindextext{%
\begin{abstract}
  In this project, we are interested in understanding the impact of a set of artifacts on the quality of video, their relationship with the content and to determine their impact on visual quality perception. The set of artifacts used in this study was chosen among those most perceptually relevant for digital video applications (e.g. blockiness, blurriness, packet-loss etc). We first plan to perform a series of psychophysical experiments on visual quality perception. These experiments will determine the impact on quality first of a set of spatial and temporal artifacts by themselves, and later of their combinations, taking into account different content with varying spatial and temporal characteristics.
\end{abstract}

% Note that keywords are not normally used for peerreview papers.
\begin{IEEEkeywords}
  Quality of video, visual quality perception, blockiness, blurriness and packet loss.
\end{IEEEkeywords}}

% make the title area
\maketitle

% To allow for easy dual compilation without having to reenter the
% abstract/keywords data, the \IEEEtitleabstractindextext text will
% not be used in maketitle, but will appear (i.e., to be "transported")
% here as \IEEEdisplaynontitleabstractindextext when the compsoc
% or transmag modes are not selected <OR> if conference mode is selected
% - because all conference papers position the abstract like regular
% papers do.
\IEEEdisplaynontitleabstractindextext
% \IEEEdisplaynontitleabstractindextext has no effect when using
% compsoc or transmag under a non-conference mode.

% For peer review papers, you can put extra information on the cover
% page as needed:
% \ifCLASSOPTIONpeerreview
% \begin{center} \bfseries EDICS Category: 3-BBND \end{center}
% \fi
%
% For peerreview papers, this IEEEtran command inserts a page break and
% creates the second title. It will be ignored for other modes.
\IEEEpeerreviewmaketitle

\section{Introduction}
% The very first letter is a 2 line initial drop letter followed
% by the rest of the first word in caps.
%
% form to use if the first word consists of a single letter:
% \IEEEPARstart{A}{demo} file is ....
%
% form to use if you need the single drop letter followed by
% normal text (unknown if ever used by IEEE):
% \IEEEPARstart{A}{}demo file is ....
%
% Some journals put the first two words in caps:
% \IEEEPARstart{T}{his demo} file is ....
%
% Here we have the typical use of a "T" for an initial drop letter
% and "HIS" in caps to complete the first word.
%%%%\IEEEPARstart{T}{his} demo file is intended to serve as a ``starter file''
%%%%for IEEE journal papers produced under \LaTeX\ using
%%%%IEEEtran.cls version 1.8 and later.

% You must have at least 2 lines in the paragraph with the drop letter
% (should never be an issue)
\IEEEPARstart{O}{ne} approach to design video quality metrics is to use artifact metrics. They have several advantages, for example simplicity and the ability to provide a good description of what is wrong (or right) with the video. But, in order to design good artifact metrics or any quality metric, it is necessary to have a good understanding of how artifacts affect quality. Therefore, it is important to study the characteristics of the most relevant artifacts present in digital video, what includes characteristics of annoyance, visibility, interactions among artifacts and content dependence. This way, we can use this data together with the data gathered previous to determine annoyance, visibility, and interaction for a diverse content.

In this experiment, we focus on $3$ particular artifacts that can be considered among the most relevant for digital video: blurriness, blockiness, and packet loss. We present these artifacts at several strengths, in several combinations and for seven original videos. In summary, we are interested in understanding "how" artifacts combine to determine annoyance and degradation in order to design better metrics.

\subsection{\textbf{Previous Experiments}}
\label{subsec:previousExp}


\subsubsection{Experiment 1}
In Experiment $1$ (\emph{Exp. $1$}), all videos were encoded using the $H.264\/AVC$ codec at a very high bitrate (approximately $120$ Mbps), with $8$ packets per frame. The coding process generated three sequences per video, with a Group of Pictures (GOP) size set to $4$, $8$ and $12$, respectively. Packet loss artifacts were then generated by dropping data packets from the bitstream~\cite{ISCAS13}. The packet loss ratios for all videos were $0.7\%$, $2.6\%$, $4.3\%$ and $8.1\%$. In total, it was generated $7$ videos x $3$ GOP x $4$ packet loss ratios = $84$ sequences. The $7$ pristine videos were added to the set, making a total of $13$ different settings per original video and resulting in $91$ test sequences~\cite{ISCAS13}.
\\
\subsubsection{Experiment 2}
Experiment $2$ (\emph{Exp. $2$}) is composed of two artifacts (blockiness and blurriness) considered the most salient present in digital videos~\cite{ICASSP05}. This experiment follows the ITU-T Recommendation P.$930$ that suggests to introduce blockiness only in regions where these artifacts are more visible. Thus, blockiness artifacts were produced by using the difference between the average of each $8 \times 8$ block and the average of the $24 \times 24$ surrounding area (which has the current $8 \times 8$ block as a center) to make each block stand out. Recommendation P.$930$ suggests the generation of blurriness with the use of a low-pass filter~\cite{ICASSP05}. Thus, blurriness artifacts were generated by applying a symmetric, two dimensional finite duration impulse response (FIR) low-pass filter to the digital image array. Different filters with varying cut-off frequencies could be used to control the amount of blurriness introduced. The blockiness and blurriness strengths for all videos were $0.4$, $0.6$ and $0.8$. In total, it was generated $7$ videos $\times$ $10$ blockiness and blurriness strengths = $70$ sequences. The $7$ pristine videos were added to the set, resulting in $77$ test sequences~\cite{SPIE03}.
\\
\subsubsection{Experiment 3}
In Experiment $3$ (\emph{Exp. $3$}) has been considered all original videos of the previous experiments. Also, it is composed of the three artifacts has been used in previous experiments (i.e., blockiness, blurriness and packet loss). In Exp. $3$ we are interested to understanding the impact on perceived quality of combinations of these artifacts and their relationship with the content. In total, we used $140$ test sequences and, details are given in the sub-section~\ref{subsec:Artifacts}.

\section{Experimental Setup}

\subsection{\textbf{Video Database}}
\label{subsec:VideoDatabase}
The video database used in this experiment was generated from seven high-definition videos (the same videos have been used in previous experiments). All them with $1280\times720$, $50$ fps and $10$ seconds duration. Representative frames of these originals are shown in Figure~\ref{fig:FirstFrame}.
\begin{figure}[h]
  \centering
  \includegraphics[width=3.5in]{../../Protocol/images/FirstFrames.jpg}
  \caption{First frame of the sequences included in Experiment.}
  \label{fig:FirstFrame}
\end{figure}

\subsection{\textbf{Artifacts}}
\label{subsec:Artifacts}
New versions of each original video were generated using different combinations of artifacts. To generate test sequences, we generate blockiness, blurriness and packet loss artifacts and combine these artifacts at different proportions and strengths. In summary, the following parameters are used to generate the test sequences:

\begin{itemize}
    \item $7$ original videos Basketball, Romeo and Juliet, Park Run, Cactus, Park Joy, Barbecue and Into Trees;
    \item $4$ different strengths of blockiness and blurriness: $0.4$, $0.6$ and $0.8$;
    \item $3$ different packet loss ratios: $0.7\%$, $2.6\%$, $4.3\%$ and $8.1\%$.
\end{itemize}

Due to a large number of combined test videos (over $200$ videos) makes it impossible to perform an experiment in a reasonable amount of time. Therefore, a reduction of the number of test sequences was achieved by selecting a subset of the original group. In this process, we eliminated most of the sequences already used in the two previous experiment already performed. We also reduced the number of parameters considering the results of the previous experiments. In summary, to select the test sequences we considered the following:

\begin{enumerate}
    \item Total time necessary to execute the overall experiment;
    \item The results of two previous experiments showed that the mean annoyance and the mean strength scores are ​​significantly different for packet loss ratios of $0.7$ and $8.1$.
    \item Among the blockiness and blurriness strength values considered in the previous experiments, we chose only the strength $0.4$ and $0.6$, which were considered to be more representative of these artifacts.
\end{enumerate}

\begin{table}[!htpb]
  \caption{Combinations of the packet loss ratio ($0.7$ and $8.1$) with blockiness and blurriness (strengths $0.4$ and $0.6$) artifacts.}
  \label{tab:PLBlockBlur}
  \centering
  % redefinindo o espaçamento das colunas
  \begin{tabular}{|c|c|c|c|}\hline
    Combination  & Packet-Loss & Blocky & Blurry  \\ \hline
            $1$  &   $0.0$     & $0.6$  & $0.0$   \\ \hline
            $2$  &   $0.0$     & $0.0$  & $0.6$   \\ \hline
            $3$  &   $8.1$     & $0.0$  & $0.0$   \\ \hline
            $4$  &   $0.7$     & $0.0$  & $0.4$   \\ \hline
            $5$  &   $8.1$     & $0.0$  & $0.4$   \\ \hline
            $6$  &   $0.7$     & $0.0$  & $0.6$   \\ \hline
            $7$  &   $8.1$     & $0.0$  & $0.6$   \\ \hline
            $8$  &   $0.7$     & $0.4$  & $0.0$   \\ \hline
            $9$  &   $8.1$     & $0.4$  & $0.0$   \\ \hline
           $10$  &   $0.7$     & $0.4$  & $0.4$   \\ \hline
           $11$  &   $8.1$     & $0.4$  & $0.4$   \\ \hline
           $12$  &   $0.7$     & $0.4$  & $0.6$   \\ \hline
           $13$  &   $8.1$     & $0.4$  & $0.6$   \\ \hline
           $14$  &   $0.7$     & $0.6$  & $0.0$   \\ \hline
           $15$  &   $8.1$     & $0.6$  & $0.0$   \\ \hline
           $16$  &   $0.7$     & $0.6$  & $0.4$   \\ \hline
           $17$  &   $8.1$     & $0.6$  & $0.4$   \\ \hline
           $18$  &   $0.7$     & $0.6$  & $0.6$   \\ \hline
           $19$  &   $8.1$     & $0.6$  & $0.6$   \\ \hline
    \end{tabular}
\end{table}

Thus, taking into account these choices, they resulted into $19$ combinations (Table~\ref{tab:PLBlockBlur}) $\times ~7$ (video versions) $+ ~7$ original videos $= 140$ videos. This is a reasonable number of sequences which made it possible to run the experiment in the allocated time of $50$ minutes.

\subsection{\textbf{Methodology}}
\label{subsec:Methodology}
For all experimental session, subjects are requested to score a set of test videos with different combinations of artifacts. The number of observers was chosen in accordance with ITU Recommendations~\cite{ITU98}. A Single Stimulus setup with implicit reference~\cite{ITU98} was adopted for the task. The experiment is running in a room with constant illumination of approximately $70$ lux. Each subject watches the stimuli (test sequences) on a $23$ inches LED monitor of resolution $1360\times768$. The distance between the subject's eyes and the video monitor is $3$ times the monitor screen's height.

Subjects are seated straight ahead of the monitor, centered at or slightly  below the eye height for most subjects. A chinrest is used to guarantee a constant distance between the subject's eyes and the monitor. A SensoMotoric Instruments iView RED Eye Tracker was used throughout the experiment to record the eye-movements of the participants (The eye tracker had a sampling rate of $50/60$Hz, a pupil tracking resolution of $0.1^{\circ}$ and a gaze position accuracy of $0.5 - 1^{\circ}$). The user interface for the experiment is implemented using the \emph{Neurobehavioral Systems} software Presentation.

All subjects are asked to watch the test sequences and indicate whether they perceived any impairment in the videos; if so, they were asked to enter how annoying the artifacts were on a continuous annoyance scale ranging between $0$ to $100$. During the experiment, the experimenter guides the subject through all sessions (calibration, free viewing, training, practice and a main experiment). Table~\ref{tab:Exp} show the summary of the settings of each experiment.

\begin{table}[!htpb]
    % increase table row spacing, adjust to taste \renewcommand{\arraystretch}{1.3}
    % if using array.sty, it might be a good idea to tweak the value of
    %\extrarowheight as needed to properly center the text within the cells
    \caption{Summary overall experiments}
    \label{tab:Exp}
    \centering
    % Some packages, such as MDW tools, offer better commands for making tables
    % than the plain LaTeX2e tabular which is used here.
    \begin{tabular}{|c||c|c|c|}
    \hline
             Settings     & Exp. $1$   & Exp. $2$   & Exp. $3$   \\  \hline
Numbers of Test Sequences &   $91$     &   $77$     &  $140$     \\  \hline
      Numbers of Subjects &   $15$     &   $16$     &  $23$      \\  \hline
            Calibration   &   $-$      & \emph{Yes} & \emph{Yes} \\  \hline
           Free Viewing   & \emph{Yes} & \emph{Yes} & \emph{Yes} \\  \hline
               Training   & \emph{Yes} & \emph{Yes} & \emph{Yes} \\  \hline
        Practice Trials   & \emph{Yes} & \emph{Yes} & \emph{Yes} \\  \hline
        Main Experiment   & \emph{Yes} & \emph{Yes} & \emph{Yes} \\  \hline
            Packet Loss   & \emph{Yes} &   $-$      & \emph{Yes} \\  \hline
             Blockiness   &   $-$      & \emph{Yes} & \emph{Yes} \\  \hline
             Blurriness   &   $-$      & \emph{Yes} & \emph{Yes} \\  \hline
    \end{tabular}
\end{table}

% An example of a floating figure using the graphicx package.
% Note that \label must occur AFTER (or within) \caption.
% For figures, \caption should occur after the \includegraphics.
% Note that IEEEtran v1.7 and later has special internal code that
% is designed to preserve the operation of \label within \caption
% even when the captionsoff option is in effect. However, because
% of issues like this, it may be the safest practice to put all your
% \label just after \caption rather than within \caption{}.
%
% Reminder: the "draftcls" or "draftclsnofoot", not "draft", class
% option should be used if it is desired that the figures are to be
% displayed while in draft mode.
%
%\begin{figure}[!t]
%\centering
%\includegraphics[width=2.5in]{myfigure}
% where an .eps filename suffix will be assumed under latex,
% and a .pdf suffix will be assumed for pdflatex; or what has been declared
% via \DeclareGraphicsExtensions.
%\caption{Simulation Results.}
%\label{fig_sim}
%\end{figure}

% Note that IEEE typically puts floats only at the top, even when this
% results in a large percentage of a column being occupied by floats.


% An example of a double column floating figure using two subfigures.
% (The subfig.sty package must be loaded for this to work.)
% The subfigure \label commands are set within each subfloat command,
% and the \label for the overall figure must come after \caption.
% \hfil is used as a separator to get equal spacing.
% Watch out that the combined width of all the subfigures on a
% line do not exceed the text width or a line break will occur.
%
%\begin{figure*}[!t]
%\centering
%\subfloat[Case I]{\includegraphics[width=2.5in]{box}%
%\label{fig_first_case}}
%\hfil
%\subfloat[Case II]{\includegraphics[width=2.5in]{box}%
%\label{fig_second_case}}
%\caption{Simulation results.}
%\label{fig_sim}
%\end{figure*}
%
% Note that often IEEE papers with subfigures do not employ subfigure
% captions (using the optional argument to \subfloat[]), but instead will
% reference/describe all of them (a), (b), etc., within the main caption.


% An example of a floating table. Note that, for IEEE style tables, the
% \caption command should come BEFORE the table. Table text will default to
% \footnotesize as IEEE normally uses this smaller font for tables.
% The \label must come after \caption as always.
%
%\begin{table}[!t]
%% increase table row spacing, adjust to taste
%\renewcommand{\arraystretch}{1.3}
% if using array.sty, it might be a good idea to tweak the value of
% \extrarowheight as needed to properly center the text within the cells
%\caption{An Example of a Table}
%\label{table_example}
%\centering
%% Some packages, such as MDW tools, offer better commands for making tables
%% than the plain LaTeX2e tabular which is used here.
%\begin{tabular}{|c||c|}
%\hline
%One & Two\\
%\hline
%Three & Four\\
%\hline
%\end{tabular}
%\end{table}

% Note that IEEE does not put floats in the very first column - or typically
% anywhere on the first page for that matter. Also, in-text middle ("here")
% positioning is not used. Most IEEE journals use top floats exclusively.
% Note that, LaTeX2e, unlike IEEE journals, places footnotes above bottom
% floats. This can be corrected via the \fnbelowfloat command of the
% stfloats package.

\section{Data Analysis}
We used the standard methods~\cite{ITU98} for analyzing the annoyance judgments provided by the test subjects. We first computed Mean Opinion Score (MOS) for each test sequence. MOS is calculated by averaging the annoyance levels over all observers for each video. In this report, values of MOS are treated as \emph{MOS1}, \emph{MOS2} and \emph{MOS3} (i.e., MOS of the \emph{Exp.1}, \emph{Exp.2} and \emph{Exp.3}, respectively). To test the relationship among the MOS values we used the Pearson correlation ($r$). To determine the significant effects between MOS's values we used a non-parametric T-Test. We also are analysing only where settings (artifacts) are the same among the experiments:
 \begin{itemize}
   \item Case $1$: from Exp. $1$ until $3$ considering only original videos;
   \item Case $2$: between Exp. $1$ and Exp. $3$, considering only the $8.1\%$ ratio of packet loss artefact;
   \item Case $3$: between Exp. $2$ and Exp. $3$, considering only the $0.6$ strengths of blockiness and blurriness artifacts (isolated).
 \end{itemize}

\subsection{\textbf{Case $1$}}
\label{subsec:case1}
We report the MOS's computed across all $7$ versions of video and all experiments, as shown in Figure~\ref{fig:AllMOS}. It is interesting to notice how the MOS' changes among the experiments: the MOS of the experiments $1$ until $3$ are $1.43$, $4.92$ and $0.80$, respectively. The Pearson linear correlation between $MOS1$ and $MOS3$ was $0.382$, $-0.0.26$ between $MOS1$ and $MOS2$, and $0.008$ between $MOS2$ and $MOS3$ for original test sequences.
\begin{figure}[h]
  \centering
  \includegraphics[width=3.7in]{MOS---AllExperiments.jpg}
  \caption{Mean Annoyance Values averaged across all the original versions of each video.}
  \label{fig:AllMOS}
\end{figure}

\subsection{\textbf{Case $2$}}
\label{subsec:case2}
In this case, we report the $MOS1$ and $MOS3$ computed to packet loss artefact across all $7$ versions of video, as shown in Figure~\ref{fig:pckMOS}. It is possible to notice that $MOS1 = 72.85$ is very higher than $MOS3 = 37.99$ with $r=.536$ and $p<.05$.
\begin{figure}[h]
  \centering
  \includegraphics[width=3.7in]{MOS---pckExp.jpg}
  \caption{Mean Annoyance Values averaged across of the packet loss versions of each video.}
  \label{fig:pckMOS}
\end{figure}

\subsection{\textbf{Case $3$}}
\label{subsec:case3}
Two different analyses were made between $MOS2$ and $MOS3$ across all $7$ versions of video: first, we report the MOS's computed only to blockiness artefact and next, we report the MOS's computed only to blurriness artefact. Figure~\ref{fig:bloMOS} shows a slightly difference between $MOS2$ and $MOS3$ to Park Joy, Into Tree and Park Run videos, but this difference increased when to Romeo \& Juliet, Cactus, Basketball and Barbecue videos. Spearman's correlation coefficient has showed that there is a moderate and positive correlation ($\rho = .536$) between them. Although, there are statistically significant different only to Romeo \& Juliet, Cactus and Basketball videos ($p<.05$).
\begin{figure}[h]
  \centering
  \includegraphics[width=3.7in]{MOS---bloExp.jpg}
  \caption{Mean Annoyance Values averaged across of the blockiness versions of each video.}
  \label{fig:bloMOS}
\end{figure}

Next, it is considered blurriness artefact. As shown in Figure~\ref{fig:bluMOS} there is a slightly difference between $MOS2$ and $MOS3$ to Park Joy, Into Tree, Cactus, Basketball and Barbecue videos increasing this different to Park Run and Romeo \& Juliet videos. Despite of strong correlation ($\rho = .857$) between them, there was not statistically significant different ($p>.05$).
\begin{figure}[h]
  \centering
  \includegraphics[width=3.7in]{MOS---bluExp.jpg}
  \caption{Mean Annoyance Values averaged across of the blurriness versions of each video.}
  \label{fig:bluMOS}
\end{figure}

\section{Data Analysis across all Experiments}
During the VARIUM tests~\cite{VARIUM13} were produced $3$ data sets spread nearly equally across settings where each experiments took into account different kinds of artifacts. Each experiment was defined by its artefact used. Specifics of the testing, such as the $H.264$ encoding parameters, can be found in~\cite{ISCAS13,ICME03,SPIE03}. The subjective MOS scores were tested by correlation the results across the experiments: a set of video sequences was common across the experiments in test. Spearman correlations of each experiment group were $.89$, $.83$ and $.87$ between $Exp.1$ and $Exp.3$ (packet loss), $Exp.2$ and $Exp.3$ (blockiness) and $Exp.2$ and $Exp.3$ (blurriness), respectively. One Spearman correlation of $.91$ was found between $Exp.2$ and $Exp.3$ (blockiness and blurriness).




\section{Conclusion}
The conclusion goes here.



% if have a single appendix:
%\appendix[Proof of the Zonklar Equations]
% or
%\appendix  % for no appendix heading
% do not use \section anymore after \appendix, only \section* is possibly needed

% use appendices with more than one appendix then use \section to start each appendix you must declare a \section before using any \subsection or using \label (\appendices by itself starts a section numbered zero.)

%%%%\appendices
%%%%\section{Proof of the First Zonklar Equation}
%%%%Appendix one text goes here.

% you can choose not to have a title for an appendix
% if you want by leaving the argument blank
%%%%\section{}
%%%%Appendix two text goes here.


% use section* for acknowledgement
%%%%\section*{Acknowledgment}
%%%%The authors would like to thank...


% Can use something like this to put references on a page
% by themselves when using endfloat and the captionsoff option.
%%%%\ifCLASSOPTIONcaptionsoff
%%%%  \newpage
%%%%\fi



% trigger a \newpage just before the given reference
% number - used to balance the columns on the last page
% adjust value as needed - may need to be readjusted if
% the document is modified later
%\IEEEtriggeratref{8}
% The "triggered" command can be changed if desired:
%\IEEEtriggercmd{\enlargethispage{-5in}}

% references section

% can use a bibliography generated by BibTeX as a .bbl file
% BibTeX documentation can be easily obtained at:
% http://www.ctan.org/tex-archive/biblio/bibtex/contrib/doc/
% The IEEEtran BibTeX style support page is at:
% http://www.michaelshell.org/tex/ieeetran/bibtex/
%\bibliographystyle{IEEEtran}
% argument is your BibTeX string definitions and bibliography database(s)
%\bibliography{IEEEabrv,../bib/paper}
%
% <OR> manually copy in the resultant .bbl file
% set second argument of \begin to the number of references
% (used to reserve space for the reference number labels box)

% biography section
%
% If you have an EPS/PDF photo (graphicx package needed) extra braces are
% needed around the contents of the optional argument to biography to prevent
% the LaTeX parser from getting confused when it sees the complicated
% \includegraphics command within an optional argument. (You could create
% your own custom macro containing the \includegraphics command to make things
% simpler here.)
%\begin{IEEEbiography}[{\includegraphics[width=1in,height=1.25in,clip,keepaspectratio]{mshell}}]{Michael Shell}
% or if you just want to reserve a space for a photo:

%%%%\begin{IEEEbiography}{Michael Shell}
%%%%Biography text here.
%%%%\end{IEEEbiography}

% if you will not have a photo at all:
%%%%\begin{IEEEbiographynophoto}{John Doe}
%%%%Biography text here.
%%%%\end{IEEEbiographynophoto}

% insert where needed to balance the two columns on the last page with
% biographies
%\newpage

%%%%\begin{IEEEbiographynophoto}{Jane Doe}
%%%%Biography text here.
%%%%\end{IEEEbiographynophoto}

% You can push biographies down or up by placing
% a \vfill before or after them. The appropriate
% use of \vfill depends on what kind of text is
% on the last page and whether or not the columns
% are being equalized.

%\vfill

% Can be used to pull up biographies so that the bottom of the last one
% is flush with the other column.
%\enlargethispage{-5in}

% that's all folks
\end{document} 